Reviewer #1:

**Scientific Quality:** Grade C (Good)

**Language Quality:** Grade B (Minor language polishing)

**Conclusion:** Accept (General priority)

**Specific Comments to Authors:** Major comments Authors presented a case in which the penile giant condyloma is accompanied with prepuce perforation. The histological study presented in Fig. 3 showed moderate dysplasia as pointed out by the co-authors. However, the inflammation and infection were not investigated, therefore it would be interesting to examine the immune cells penetration in the lesion area compared to that from healthy region. Moreover, the inflammation could be investigated via examining the expression levels of interleukins (ILs). Minor comments In Fig. 3, in the microscopic image, scale bar to be added

Response: At a magnification of 40x in the microscopic images, we observed a significant presence of large immune infiltrates within the dermis distal to the lesion, in comparison to the healthy region. I apologize that we did not retain fresh specimens for further examination of interleukin levels. The scale bar in the microscopic images is set at 20-fold magnification (hematoxylin and eosin staining, × 20).

Reviewer #2:

**Scientific Quality:** Grade C (Good)

**Language Quality:** Grade B (Minor language polishing)

**Conclusion:** Major revision

**Specific Comments to Authors:** HPV genotyping was performed in this patient? All giant condiloma acuminatum (CA) are not Buschke-Lowenstein tumor. more information is needed. In this case, only multiple CA are seen. discussion section should be improved. the authors can add the studies below to increase the value of case report. Sarier M, Ceyhan AM, Sepin N, et al. HPV infection in urology practice. Int Urol Nephrol. October 2019:1-8. doi:10.1007/s11255-019-02302-2 Sarier M, Sepin N, Emek M, Konuk EY, Kaplan T, Yuksel BA, Barut Z, Cakcak DS, Hoscan MB. Evaluation of the optimal sampling approach for HPV genotyping in circumcised heterosexual men with genital warts. J Infect Chemother. 2023 May;29(5):475-480. doi: 10.1016/j.jiac.2023.01.017

Response: The HPV genotyping results for the redundant prepuce specimen
indicated a positive presence of HPV-11. The pathology confirmed the
diagnosis of giant condyloma acuminata with moderate dysplasia and
destructive local invasion without metastasizing. I have read and cited the
references you recommended. I would like to express my gratitude to the
reviewer for providing a comprehensive and informative review, which has
allowed me to gain additional knowledge. As a result, I have been able to
enhance and refine this case report.